Author's response to reviews

Title: Objective assessment of dental fluorosis in an epidemiological setting

Authors:

Iain A Pretty (iain.pretty@manchester.ac.uk)
Michael McGrady (michael.mcgrady@manchester.ac.uk)
Christian Zakian (christian.zakian@manchester.ac.uk)
Roger P Ellwood (roger.ellwood@manchester.ac.uk)
Andrew Taylor (andrew.taylor@manchester.ac.uk)
Mohammed O Sharif (owaise2000@hotmail.com)
Timothy Iafolla (Timothy.Iafolla@nih.hhs.gov)
Angeles Martinze-Mier (esmartin@iupui.edu)
Patcharawan Srisilapanan (patcharawana@gmail.com)
Narumanas Korwanich (dnncmi002@chiangmai.ac.th)
Michaela Goodwin (Michaela.michaela.goodwin@manchester.ac.uk)
Bruce A Dye (bfd1@cdc.gov)

Version: 5 Date: 24 April 2012

Author's response to reviews: see over
IP/ms

24-Apr-12

Prof Ratilal RL Lalloo
BMC Public Health
Editorial Office

Dear Prof Lalloo

Re: MS: 2073542652631854

Thank you for sending the comments of the two reviewers in relation to the above captioned manuscript.

I can confirm that we have:

a) Corrected and extended the list of abbreviations

b) Uploaded two composite figures. We have uploaded Figure 4 and 5 as labelled composites and Figure 6 and 7 are the same images without the labels

I do hope that this is satisfactory but do please contact me if you require anything further.

Kindest regards

IAIN A PRETTY